Author's response to reviews

Title: MicroRNAs in urine are not biomarkers of multiple myeloma

Authors:

Lenka Sedlarikova (lena.sedlarikova@gmail.com)
Lenka Besse (lenka.kubiczek@gmail.com)
Sona Novosadova (357892@mail.muni.cz)
Veronika Kubaczkova (veronika.kubaczkova@seznam.cz)
Lenka Radova (avodar@gmail.com)
Michal Stanik (stanik@mou.cz)
Marta Krejci (mkrejci@fnbrno.cz)
Roman Hajek (rom.hajek@seznam.cz)
Sabina Sevcikova (sevcik@med.muni.cz)

Version: 6
Date: 14 September 2015

Author's response to reviews:

Dear Editor,

Please, find enclosed the corrected version of our manuscript. We have changed the order of individual parts of the manuscript to better follow the guidelines of the journal.

As for the question of the reviewer, in Table 1- there are 22 healthy donors, we rechecked the protocols and changed the table accordingly (line 2: males-females).

We do apologize for this delay, and we hope that now the manuscript is ready for acceptance.

Sincerely,

Sabina Sevcikova